



Geophysical Research Letters

RESEARCH LETTER

10.1029/2018GL078888

Kev Points:

- The focus of climate mitigation policy on singular thresholds of global mean temperature conceals significant variability in regional impacts
- We propose a simple tool which maps how much warming is needed in one location to match the effects of reaching a specific warming threshold in another
- Expanding the "temperature of equivalence" framework could help with integrated adaptation planning in a warmer world

Supporting Information:

• Supporting Information S1

Correspondence to:

L. J. Harrington, luke.harrington@ouce.ox.ac.uk

Citation:

Harrington, L. J., Frame, D., King, A. D., & Otto, F. E. L. (2018). How uneven are changes to impact-relevant climate hazards in a 1.5 °C world and beyond? Geophysical Research Letters, 45, 6672-6680. https://doi.org/10.1029/ 2018GL078888

Received 19 APR 2018 Accepted 14 JUN 2018 Accepted article online 25 JUN 2018 Published online 13 JUL 2018

The copyright line for this article was changed on 21 JUL 2018 after original online publication.

Luke J. Harrington¹, Dave Frame², Andrew D. King³, and Friederike E. L. Otto¹ ¹Environmental Change Institute, University of Oxford, Oxford, UK, ²New Zealand Climate Change Research Institute, Victoria University of Wellington, Wellington, New Zealand, ³ARC Centre of Excellence for Climate System Science, School of

How Uneven Are Changes to Impact-Relevant Climate Hazards

in a 1.5 °C World and Beyond?

Earth Sciences, University of Melbourne, Melbourne, Victoria, Australia

Abstract In the last decade, climate mitigation policy has galvanized around staying below specified thresholds of global mean temperature, with an understanding that exceeding these thresholds may result in dangerous interference of the climate system. United Nations Framework Convention on Climate Change texts have developed thresholds in which the aim is to limit warming to well below 2 °C of warming above preindustrial levels, with an additional aspirational target of 1.5 °C. However, denoting a specific threshold of global mean temperatures as a target for avoiding damaging climate impacts implicitly obscures potentially significant regional variations in the magnitude of these projected impacts. This study introduces a simple framework to quantify the magnitude of this heterogeneity in changing climate hazards at 1.5 °C of warming, using case studies of emergent increases in temperature and rainfall extremes. For example, we find that up to double the amount of global warming (3.0 °C) is needed before people in high-income countries experience the same relative changes in extreme heat that low-income nations should anticipate after only 1.5 °C of warming. By mapping how much warming is needed in one location to match the impacts of a fixed temperature threshold in another location, this "temperature of equivalence" index is a flexible and easy-to-understand communication tool, with the potential to inform where targeted support for adaptation projects should be prioritized in a warming world.

Plain Language Summary While the threshold of global mean warming which results in damaging climate impacts is thought to differ significantly between different locations, quantifying these differences has proven difficult for the scientific community. This paper introduces a simple tool, called the "temperature of equivalence" index, which maps how much global warming is needed for one location to match the effects of reaching a specific warming threshold in another. As an illustration of the framework, we find changes to the severity of extreme heat events for low-income nations after 1.5 °C of warming would not be seen for other regions of the world until after a global temperature rise twice as high. By aggregating the temperature of equivalence index for previously incompatible measures of the impacts of climate change, future work could enable a more holistic understanding of which nations will have shared experiences in a warmer world, with potential benefits for adaptation planning as a consequence.

1. Introduction

There is significant societal interest in understanding how climate change impacts will manifest themselves in response to continued increases in global mean temperature (Seneviratne et al., 2016; Stott, 2016; Stott et al., 2016). In the context of the development of future mitigation targets (Schleussner, Rogelj, et al., 2016), there is also an enhanced need to more specifically understand how regional changes to weather extremes and other climate-related hazards will manifest themselves in response to specific thresholds of global mean warming (Hawkins et al., 2017; King et al., 2017; Mitchell et al., 2016). Following the signing of the Paris Agreement in December 2015, there has been a new focus on the regional impacts associated with "1.5° of warming above preindustrial levels," not previously the focus of targeted scientific research (Mitchell et al., 2016; Rogelj & Knutti, 2016; Schleussner, Lissner, et al., 2016). Of course, the question of whether the exceedance of any specific threshold of global-mean warming translates to "dangerous interference" of the climate system inherently simplifies potentially significant regional differences in the severity of such impacts—or more precisely, the threshold of warming which would result in damaging local impacts may differ significantly between different locations. However, resolving how different these "thresholds" are remains poorly constrained and is thus the focus of this analysis.

©2018. American Geophysical Union. All Rights Reserved.

One method which has proven useful to characterize relative changes to future climate between different locations around the world has been that of "spatial climate analogues." Primarily used as a communication tool (for example, http://bit.ly/2n1ubbF), this approach commonly considers what regions of the world might today exhibit climatic conditions comparable to the projections expected for a different location under a future warming scenario. Many papers have presented variations on this theme (Brown & Katz, 1995; Hallegatte et al., 2007; Kalkstein & Greene, 1997; Pugh et al., 2016; Williams et al., 2007), each with different caveats: The most recent example from Dahinden et al. (2017) found that up to 20% of today's local climates, when characterized based on seasonal cycles of average temperature and precipitation over land regions, would disappear following an additional 2.0 °C rise in global temperatures.

This study presents a new framework motivated by the concept of spatial climate analogues, but with a focus on assessing the level of *dissimilarity* in regional climate hazards following a given rise in global mean temperatures. Specifically, a climate model-based framework is used to identify how many degrees of global warming would be required for each individual grid cell around the world to reach the same threshold of emergent climate change as those experienced at 1.5 °C of warming for a specific location of interest. While nearly all previous approaches to spatial climate analogues keep the present-day climate fixed as a frame of reference, this paper's approach focuses on future warming thresholds as the target reference frame, thereby enabling a clearer demonstration of regional variability in *future* climate change and related policy implications.

Using emergent increases in the frequency of heat and precipitation extremes as examples (sections 2 and 3), section 4 will then discuss how an extended framework of comparing the emergence of climate change impacts above background variability across regions can have clear benefits for decision makers worldwide. Such benefits include (1) demonstrating that the emergent changes expected at some point in the future with continued carbon emissions in one part of the world are in fact closely related to the changes being felt in other regions of the world today (with the potential to then learn from the experiences of those places) and (2) highlighting which regions should be the focus of more targeted support for adaptation planning, given the severity of some changes which are projected to occur even if ambitious mitigation scenarios are successfully achieved.

2. Data and Methods

For illustrative purposes, this study utilizes two extreme temperature and precipitation metrics from the Expert Team on Climate Change Detection Indices (ETCCDIs, Sillmann et al., 2013): for temperature, we analyze the annual maxima of all daily maximum temperatures (hereafter TXx); for precipitation, we identify annual maximum 1-day precipitation accumulations at each location (Rx1day).

To facilitate a like-for-like comparison of the changes expected in a multitude of different climatic variables with continued warming, we focus on signal-to-noise (S/N) ratios for each individual variable. This normalizes the magnitude of climate changes with respect to the amount of natural variability expected at a certain location in the reference climate and thereby provides a method of circumventing some difficulties that otherwise arise when comparing metrics with different definitions and different units of measurement. However, the underlying premise of identifying "temperatures of equivalence" could equally be applied for absolute changes in TXx, and the importance of examining a variety of indices is discussed further in section 4.2.

For 23 models contributing to the Coupled Model Intercomparison Project Phase 5 (CMIP5; Taylor et al., 2012) which provided sufficient data for the calculation of TXx and Rx1day (Table S1 in the supporting information), time series of S/N ratios are calculated for each metric over the period 1861–2100, using "Historical" and "RCP8.5" simulations concatenated together. Only a single ensemble member (r111p1) is selected from each model. Following previously published methods (King et al., 2015, 2016), the signal of each variable at each grid cell is calculated as the mean over a running 21-year time window, relative to the mean over the period 1861–1880. The noise of each variable is calculated for each grid cell within each model as the standard deviation over 1901–2000, using Historical model simulations detrended with a simple linear fit. The S/N ratios are calculated for each model at their native resolution first, before being interpolated to a common $2.5^{\circ} \times 2.5^{\circ}$ grid to facilitate comparative analysis later.

The corresponding estimate of global mean temperature anomalies for each model is also calculated using a 21-year running mean, and with respect to an 1861–1880 baseline. It is noted here that global-mean



Figure 1. (a) Multimodel median temperature of equivalence (TE) with respect to changes in TXx signal-to-noise ratios over land regions in India at the 21-year period corresponding to 1.5 °C of global mean warming (with respect to 1861–1880). (b) TE1.5 within each CMIP5 model, expressed as a cumulative distribution function for the global population, using high-resolution gridded data for the year 2015. The dashed red line shows 1.5 °C of warming, which is the benchmark for the changes considered in the region of interest. Inset statistics in (b) show the fraction of people which have a TE1.5 of less than 1.0 °C or more than 3.0 °C (with best-guess estimates and a full model range presented respectively).

temperatures are calculated simply as the area-weighted average of near-surface air temperatures over all land and ocean regions in each model (that is, with no masking to specific regions with high observational coverage).

Since information about climate change impacts associated with exceeding a specific threshold of global mean warming will be more desirable for decision makers (Knutti et al., 2016; Rogelj & Knutti, 2016), we next identify the 21-year period for which a global mean warming anomaly of 1.5 °C is exceeded within each individual model and remains exceeded until the end of the time series (2100). The corresponding S/N ratio for each of the two ETCCDIs is then extracted for this same 21-year period.

2.1. Defining the Temperature of Equivalence Index

To illustrate the "temperature of equivalence" (hereafter TE) index as a concept, we begin with a simple demonstration using TXx and a region roughly encompassing India as a case study. Aggregating across all land grid cells in the black rectangular box ($8^{\circ}N-35^{\circ}N$, $70^{\circ}E-95^{\circ}E$) in Figure 1a, the S/N ratio of TXx changes is found for each individual climate model at the point of 1.5 °C of warming. We then look at all other grid cells worldwide separately and assess how much warming is required within that same model to produce the same emergent increases in TXx as was found for India after 1.5 °C of warming. The map in Figure 1 demonstrates the multimodel median amount of warming required for each grid cell to experience the same changes in TXx as for India—that is, the TE with respect to changes at 1.5 °C (or TE1.5). As expected, one can see that there are equal fractions of land with a TE1.5 just below (light blue) or above (yellow) 1.5 °C within the black rectangle itself.

The right-hand panel presents another way of interpreting the spatial patterns TE1.5 for TXx relative to the region of interest, showing the cumulative distribution of TE exposure for the global population. Population data for both this and all subsequent figures have been taken for the year 2015 from the Center for International Earth Science Information Network database (CIESIN, 2016). The population data were first aggregated from the $0.25^{\circ} \times 0.25^{\circ}$ spatial resolution provided, to the $2.5^{\circ} \times 2.5^{\circ}$ resolution of the climate model output. As can be seen by the individual lines shown for each CMIP5 model in Figure 1b, as much as 20% of the global population experiences the same *emergent* changes in TXx after less than 1.0 °C of warming as is found in India after 1.5 °C, while all models show at least 65% of the global population to have experienced the same changes after 3.0 °C of warming.

The remaining examples in this study do not consider the TE index with respect to a simple rectangular box as the location of interest, but instead with respect to the aggregation of S/N ratios from all low-income countries together, using classifications taken from the World Bank (2016, http://bit.ly/2bBWnzX; Figure S10 in the supporting information). Similarly, instead of considering population exposure for the *entire globe*



Figure 2. (a) Map of the multimodel median temperature of equivalence (TE) index, when comparing against the signal-tonoise ratio of TXx expected for the average citizen of a low-income nation after 1.5 °C of warming. (b) TE1.5 for the population of all high-income nation aggregated together, expressed as a cumulative distribution function using highresolution gridded data for the year 2015. The dashed red line shows 1.5 °C of warming; the white land regions show where no "equivalent" emergence occurs by 2100 under RCP8.5; the thick blue line shows the multimodel median response; the thin blue lines indicate the 16th–84th percent model spread. Inset statistics in (b) show the fraction of people which have a TE1.5 of less than 1.0 °C or more than 3.0 °C.

as a way of presenting how the changes at 1.5 °C for one location compares with others (as with Figure 1b), we will hereafter compare against those people who live in high-income nations (Figure S10). Not only does this framework illustrate the versatility of the TE concept, but aggregating regional groups according to shared socioeconomic characteristics helps to characterize similar levels of capability in addressing the impacts of climate change (Frame et al., 2017).

2.2. Methodological Considerations

While the intention is primarily to illustrate practical examples of the TE concept, the decision to examine S/N ratios for the chosen extreme climate indices is motivated by the multitude of studies which have placed an increasing emphasis on interpreting the signal of regional climate change in the context of existing variability over a given location (Diffenbaugh & Scherer, 2011; Fischer & Knutti, 2015; Frame et al., 2017; Harrington et al., 2016; Hawkins & Sutton, 2012; King et al., 2016; Lehner & Stocker, 2015; Mahlstein et al., 2011). It is further noted that this study only calculates S/N ratios at a given warming threshold from a high-carbon scenario of future climate change (RCP8.5): Such a scenario, which crosses a wide range of temperatures, is a necessary requirement for the framing of results presented in both sections 2.1 and 3. However, previous studies have shown that, for many ETCCDIs, the response to the same increase in global temperatures is largely similar across RCP scenarios—that is, scenario uncertainty across RCPs is significantly smaller than corresponding model uncertainty (Harrington et al., 2016; Seneviratne et al., 2016; Wartenburger et al., 2017).

Finally, it is important to acknowledge that this framework could be easily applied to a variety of other temperature thresholds—the 1.5 °C example has been simply chosen here to provide relevance for decision making about the ambitious targets of the 2015 Paris Agreement. Equivalent results for 2.0 and 2.5 °C of warming are presented in the supporting information.

3. Results

3.1. Example of Monotonic Climate Changes: Temperature Extremes

Figure 2 shows the patterns of TE when considering the multimodel median change in TXx S/N ratios following 1.5 °C of global warming for all low-income nations aggregated together. As the left-hand panel reveals, there is a notable latitudinal gradient in the amount of warming required to match this level of emergent increase in heat extremes. For some lower latitudes, such changes have already been reached before 1.0 °C of warming (a threshold thought to have been already exceeded in 2017; Haustein et al., 2017), while the vast majority of continental locations in midlatitudes require between 2 and 3.5 °C of warming to reach these levels.



Figure 3. Same as for Figure 2, but showing the patterns of TE1.5 when focusing on the emergent changes in Rx1day for the average citizen of a low-income nation.

When aggregating these spatial results from Figure 2a according to where people in high-income countries live, clearer results begin to emerge. Specifically, we find in Figure 2b that the same amount of emergent changes in temperature extremes felt by low-income countries after 1.5 °C of warming would be reached instead for an equal fraction of high-income populations after 2.6 °C (1.9–3.0 °C) of warming. Another way of interpreting this result is that the changes in extreme heat felt by the average citizen of a low-income country after 1.5 °C of warming would not be felt by about 40% (16th–84th percentile range of 12%–51%) of people living in high-income nations until well after double the amount of global warming (3.0 °C) is reached. Such disparities are significant, and underscore how different the rates of emergent changes in extreme heat can be for different locations (Figure S1a). Further versions of Figure 2 can be found in supporting information for several other global temperature thresholds and using a different metric of extreme heat (TX90p)—all results show qualitatively comparable outcomes.

3.2. Impact-Relevant Climate Changes With Spatially Divergent Responses to Warming: High-Precipitation Extremes

For a climatic variable which experiences monotonic changes in response to greenhouse gas emissions, like extreme temperatures, the TE index can be well understood and interpreted in a relatively straightforward way. Some interpretation difficulties do, however, arise when applying the TE framework to a variable for which climate change may induce a decreasing or increasing signal of change, depending on the location considered. To illustrate such an example, we replicate the analysis of Figure 2, but this time using S/N ratios of annual-maximum 24-hr precipitation accumulations (hereafter Rx1day) at each grid point, instead of TXx.

First, it is important to highlight that for the majority of populated regions around the world, projections show an increase in the intensity of daily-scale rainfall extremes to emerge after 1.5 °C of warming (Figure S1b). When these changes for low-income countries are presented as a map of TE1.5 however (Figure 3a), it becomes clear that there are many locations which have already seen the same emergent changes in extreme rainfall (blue grid cells) as expected after 1.5 °C for the average citizen of a low-income country, while such changes will never occur for other locations (white land regions). These disparities are quantified more explicitly in Figure 3b: To a first-order approximation, half of all people in high-income countries experience the same amount of emergent increases in extreme rainfall after 1.0 °C as for inhabitants of low-income countries after 1.5 °C of warming, while some 20% (10%–30%) of people will never see such changes. This partly reflects the fact that several low-income countries span arid regions in Africa which show very small S/N ratios (Figures S2 and S10); hence, many Northern Hemisphere regions experience equivalent changes in extreme rainfall after much less warming.

These results are nevertheless still as informative as those for temperature extremes: People living in one location can identify other places around the world which will experience a comparable increase in the severity of rainfall extremes after some threshold of global mean warming—it just so happens that the differences in these projected signals of change are more geographically diverse than is the case for changes in extreme



heat. For example, those places in the high northern latitudes with robust projected increases in extreme rainfall frequency (Figure S1b; Min et al., 2011) would not expect to find shared future experiences with those arid regions of the world which anticipate negligible or even opposing signals of extreme precipitation changes in the future (Pfahl et al., 2017).

4. Discussion

4.1. Potential Applications of the Temperature of Equivalence Framework

The examples provided in this study demonstrate two possible applications of the TE index: (1) providing a means of comparing between two countries (or groups of countries), to determine what locations will experience similar changes in response to some threshold of continued warming and thus have shared interests in developing adaptation frameworks and (2) demonstrating how anomalous the emergent hazards for a specific community may be after exceeding some threshold of global warming, especially when compared to projections for the global population as a whole.

The TE framework also represents a powerful tool to highlight how a large fraction of the many potential hazards anticipated with climate change are actually similar for many different regions around the world, but with the important caveat that the amount of warming required for these changes to emerge will be different in different locations. The TE concept could therefore provide an effective framework to facilitate integrated climate adaptation plans: For example, every individual country could calculate, for a range of future global temperature thresholds of interest to local planners (such as 1.5, 2, 2.5 °C, or beyond), which other countries (if they exist) best represent a similar level of emergent change in the present day. This would then incentivize collaborative efforts between the two nations to prepare adaptation plans for this fast emerging country, as it would also serve as a useful waypoint for the country which is anticipating similar changing hazards in the coming decades.

4.2. Considerations for Extending the TE Framework to Other Impact-Relevant Metrics of Climate Change

This approach to integrated adaptation planning could be examined for many metrics of a changing climate for which the TE framework could be implemented. However, for the TE framework to be most successful, scientists would need to provide calculations for a wide range of indices which characterize impact-relevant climate hazards—similar to the examples of section 3—but without making any assumptions a priori about what will be most useful for stakeholders, and instead leaving those choices to downstream decision makers (Hewitt et al., 2017).

Even for the case of multiple temperature-related hazards in a warming world, the TE index for a 1.5 °C world should be calculated for a range of metrics which capture both absolute and normalized changes for both direct and wet-bulb temperatures, as each of these metrics is specifically relevant to different types of heat-related hazards. For the case of direct heat stress, Im et al. (2018) demonstrate that the fraction of time exceeding specific wet-bulb temperature thresholds of 28, 31, and 35 °C are important indices to consider, as such levels constitute upper bounds of heat-regulating capabilities for the human body (Sherwood & Huber, 2010). Equally, multiple studies of nonambient temperature mortality rates show that individual cities have optimal temperature bounds which are specific to their historical climate (Gasparrini et al., 2015; Tobías et al., 2017), thus suggesting that deviations beyond historical familiarity (or S/N ratios in temperature) will also be important indices for local stakeholders and health practitioners.

Impacts specifically assessed by the ISI-MIP project (Warszawski et al., 2014) or more targeted regional modeling projects (Giorgi & Gutowski, 2015) are also possible candidates, with examples including hydrological drought severity, how often growing season temperatures exceed optimum thresholds for specific crop types, or even changes to freshwater availability. If specifically assessed using these fit-for-purpose modeling frameworks, the potential to aggregate multiple climate change impacts under the banner of the TE index could also help to alleviate difficulties in interpreting nonlinear or sigmoidal patterns of change which might emerge with further warming (Ricke et al., 2016). By providing a method which equates the relative speed of changes to this myriad of impact-relevant hazards in a warming world under a consistent metric, the TE index could be used as a way to more accurately resolve which countries will in fact experience the fastest and most severe effects from a specific *N*-degree rise in global mean

temperatures and thus warrant targeted support from those countries which may experience relatively less severe changes.

Other types of important physical climate changes are more difficult candidates to be assessed under the TE framework: One such example is sea level rise. While Lyu et al. (2014) presented a useful "time of emergence" analysis (Hawkins & Sutton, 2012) for regional sea level changes in the future, this approach may not be the most informative in terms of comparing the relative impacts of changing coastlines in a future climate. A more appropriate metric for a TE-based analysis might instead be what fraction of a country's gross domestic product or population (Hallegatte et al., 2013) are directly exposed to a 1-in-N-year coastal inundation event with continuing warming—that way, the TE framework could then be easily used to show a like-for-like comparison between different locations worldwide.

For all of the above examples, we continue to emphasize that the specific and often value-laden (O'Brien & Wolf, 2010) selection of which metrics of climate change are most relevant for use in a TE framework should remain for local and regional decision makers to independently determine. However, by providing a broad range of metrics (which is the subject of ongoing work), each of which are directly relevant to multiple climate change impacts, we argue the TE index represents a potentially powerful tool for stakeholders.

4.3. Limitations

The framework proposed in this study has some methodological limitations which need to be carefully considered—these primarily relate to whether the transient climate change scenarios currently available to characterize TE1.5 are fit-for-purpose. Our analysis has made use of the high-emissions RCP8.5 scenario as there is a necessary requirement for the future climate trajectory to span as wide a range of temperature thresholds as possible. However, there are open questions about (1) how the impacts of an *N*-degree increase in global mean temperatures are influenced by the emissions pathway taken to reach that threshold, such as regional changes to aerosol emissions (Samset et al., 2018; Wang et al., 2017), as well as (2) whether reaching a global temperature threshold in a transient scenario will generate the same climate impacts as for a case where temperatures stabilize at that threshold (James et al., 2017). These remain open questions which could influence the TE index for some individual locations and for some types of impacts and should thus be a priority for future research.

5. Summary

The TE framework is a relatively simple and easy-to-understand method to characterize the dramatic differences in regional climate hazards projected for future increases in global temperatures. This has been applied to two examples of precipitation and temperature extremes, but we emphasize the potential value of extending this framework to a range of climate change indices of interest to regional decision makers. For our illustrative results using only one specific metric of characterizing heat extremes, even the most aggressive mitigation scenarios can result in changes to the severity of extreme heat events for low-income populations which would not be seen for other regions of the world until after a global temperature rise twice as high. By aggregating the TE index for previously incompatible measures of the impacts of climate change, future work could enable a more holistic characterization of how different regions of the world will experience the impacts of a 1.5 °C warmer world and beyond. With an improved understanding of which nations will share common experiences under future climate change, the TE framework may also help to facilitate the prospect of joint adaptation plans for different countries worldwide.

References

Brown, B. G., & Katz, R. W. (1995). Regional analysis of temperature extremes: Spatial analog for climate change? Journal of Climate, 8(1), 108–119. https://doi.org/10.1175/1520-0442(1995)008<0108:RAOTES>2.0.CO;2

Center for International Earth Science Information Network - CIESIN - Columbia University (2016). Gridded Population of the World, Version 4 (GPWv4): Population Count, Revision 9. Palisades, New York: NASA Socioeconomic Data and Applications Center (SEDAC). https://doi.org/ 10.7927/H4PG1PP

Dahinden, F., Fischer, E. M., & Knutti, R. (2017). Future local climate unlike currently observed anywhere. *Environmental Research Letters*, 12(8), 084004. https://doi.org/10.1088/1748-9326/aa75d7

Diffenbaugh, N. S., & Scherer, M. (2011). Observational and model evidence of global emergence of permanent, unprecedented heat in the 20th and 21st centuries. *Climatic Change*, 107(3–4), 615–624. https://doi.org/10.1007/s10584-011-0112-y

Fischer, E. M., & Knutti, R. (2015). Anthropogenic contribution to global occurrence of heavy-precipitation and high-temperature extremes. Nature Climate Change, 5(6), 560–564. https://doi.org/10.1038/nclimate2617

Acknowledgments

The authors acknowledge the World Climate Research Programme's Working Group on Coupled Modeling, which is responsible for CMIP, and thank the climate modeling groups for producing and making available their model output. For CMIP the U.S. Department of Energy's Program for Climate Model **Diagnosis and Intercomparison** provides coordinating support and led development of software infrastructure in partnership with the Global Organization for Earth System Science Portals. The authors would also like to thank the providers of the ETCCDI output, calculated for the different CMIP5 runs, which were freely obtained from the Environment Canada CLIMDEX website, http://cccma.ec.gc.ca/data/ climdex/. L.J.H. acknowledges support from the Transition into the Anthropocene (TITAN) project, funded by a European Research Council (ERC) Advanced Grant (EC-320691), and the MaRIUS project: Managing the Risks, Impacts and Uncertainties of droughts and water Scarcity, funded by the Natural Environment Research Council (NE/L010364/1). D.J.F. and L.J.H. acknowledge support from the New Zealand Deep South National Science Challenge, A.D.K. was supported by the Australian Research Council (CE110001028 and DE180100638).



Frame, D., Joshi, M., Hawkins, E., Harrington, L. J., & de Roiste, M. (2017). Population-based emergence of unfamiliar climates. *Nature Climate Change*, 7(6), 407–411. https://doi.org/10.1038/nclimate3297

Gasparrini, A., Guo, Y., Hashizume, M., Lavigne, E., Zanobetti, A., Schwartz, J., et al. (2015). Mortality risk attributable to high and low ambient temperature: A multicountry observational study. *The Lancet*, 386(9991), 369–375. https://doi.org/10.1016/S0140-6736(14)62114-0

Giorgi, F., & Gutowski, W. J. (2015). Regional dynamical downscaling and the CORDEX initiative. Annual Review of Environment and Resources, 40(1), 467–490. https://doi.org/10.1146/annurev-environ-102014-021217

Hallegatte, S., Green, C., Nicholls, R. J., & Corfee-Morlot, J. (2013). Future flood losses in major coastal cities. *Nature Climate Change*, 3(9), 802–806. https://doi.org/10.1038/nclimate1979

Hallegatte, S., Hourcade, J.-C., & Ambrosi, P. (2007). Using climate analogues for assessing climate change economic impacts in urban areas. *Climatic Change*, 82(1-2), 47–60. https://doi.org/10.1007/s10584-006-9161-z

Harrington, L. J., Frame, D. J., Fischer, E. M., Hawkins, E., Joshi, M., & Jones, C. D. (2016). Poorest countries experience earlier anthropogenic emergence of daily temperature extremes. *Environmental Research Letters*, 11(5), 055007. https://doi.org/10.1088/1748-9326/11/5/055007

Haustein, K., Allen, M. R., Forster, P. M., Otto, F. E. L., Mitchell, D. M., Matthews, H. D., & Frame, D. J. (2017). A real-time global warming index. Scientific Reports, 7, 15417.

Hawkins, E., Ortega, P., Suckling, E., Schurer, A., Hegerl, G., Jones, P., et al. (2017). Estimating changes in global temperature since the preindustrial period. *Bulletin of the American Meteorological Society*, *98*(9), 1841–1856. https://doi.org/10.1175/BAMS-D-16-0007.1

Hawkins, E., & Sutton, R. (2012). Time of emergence of climate signals. *Geophysical Research Letters*, 39, L01702. https://doi.org/10.1029/2011GL050087

Hewitt, C. D., Stone, R. C., & Tait, A. B. (2017). Improving the use of climate information in decision-making. *Nature Climate Change*, 7(9), 614–616. https://doi.org/10.1038/nclimate3378

Im, E.-S., Kang, S., & Eltahir, E. A. B. (2018). Projections of rising heat stress over the western Maritime Continent from dynamically downscaled climate simulations. Global and Planetary Change, 165, 160–172. https://doi.org/10.1016/j.gloplacha.2018.02.014

James, R., Washington, R., Schleussner, C.-F., Rogelj, J., & Conway, D. (2017). Characterizing half-a-degree difference: A review of methods for identifying regional climate responses to global warming targets. Wiley Interdisciplinary Reviews: Climate Change, 8, e457. https://doi.org/ 10.1002/wcc.457

Kalkstein, L. S., & Greene, J. S. (1997). An evaluation of climate/mortality relationships in large U.S. cities and the possible impacts of a climate change. *Environmental Health Perspectives*, 105(1), 84–93. https://doi.org/10.1289/ehp.9710584

King, A. D., Black, M. T., Min, S.-K., Fischer, E. M., Mitchell, D. M., Harrington, L. J., & Perkins-Kirkpatrick, S. E. (2016). Emergence of heat extremes attributable to anthropogenic influences. *Geophysical Research Letters*, 43, 3438–3443. https://doi.org/10.1002/2015GL067448

King, A. D., Donat, M. G., Fischer, E. M., Hawkins, E., Alexander, L. V., Karoly, D. J., et al. (2015). The timing of anthropogenic emergence in simulated climate extremes. *Environmental Research Letters*, 10(9), 094015. https://doi.org/10.1088/1748-9326/10/9/094015

King, A. D., Karoly, D. J., & Henley, B. J. (2017). Australian climate extremes at 1.5 °C and 2 °C of global warming. Nature Climate Change, 7, 412–416.

Knutti, R., Rogelj, J., Sedláček, J., & Fischer, E. M. (2016). A scientific critique of the two-degree climate change target. Nature Geoscience, 9(1), 13–18. https://doi.org/10.1038/ngeo2595

Lehner, F., & Stocker, T. F. (2015). From local perception to global perspective. Nature Climate Change, 5(8), 731–734. https://doi.org/10.1038/ nclimate2660

Lyu, K., Zhang, X., Church, J. A., Slangen, A. B. A., & Hu, J. (2014). Time of emergence for regional sea-level change. *Nature Climate Change*, 4(11), 1006–1010. https://doi.org/10.1038/nclimate2397

Mahlstein, I., Knutti, R., Solomon, S., & Portmann, R. W. (2011). Early onset of significant local warming in low latitude countries. *Environmental Research Letters*, 6(3), 034009. https://doi.org/10.1088/1748-9326/6/3/034009

Min, S.-K., Zhang, X., Zwiers, F. W., & Hegerl, G. C. (2011). Human contribution to more-intense precipitation extremes. *Nature*, 470(7334), 378–381. https://doi.org/10.1038/nature09763

Mitchell, D., James, R., Forster, P. M., Betts, R. A., Shiogama, H., & Allen, M. (2016). Realizing the impacts of a 1.5 °C warmer world. *Nature Climate Change*, 6(8), 735–737. https://doi.org/10.1038/nclimate3055

O'Brien, K., & Wolf, J. (2010). A values-based approach to vulnerability and adaptation to climate change. Wiley Interdisciplinary Reviews: Climate Change, 1, 232–242.

Pfahl, S., O'Gorman, P. A., & Fischer, E. M. (2017). Understanding the regional pattern of projected future changes in extreme precipitation. *Nature Climate Change*, 7, 423–427.

Pugh, T. A. M., Müller, C., Elliott, J., Deryng, D., Folberth, C., Olin, S., et al. (2016). Climate analogues suggest limited potential for intensification of production on current croplands under climate change. *Nature Communications*, *7*, 12608. https://doi.org/10.1038/ncomms1260

Ricke, K. L., Moreno-Cruz, J. B., Schewe, J., Levermann, A., & Caldeira, K. (2016). Policy thresholds in mitigation. Nature Geoscience, 9(1), 5–6. https://doi.org/10.1038/ngeo2607

Rogelj, J., & Knutti, R. (2016). Geosciences after Paris. Nature Geoscience, 9, 187-189.

Samset, B. H., Sand, M., Smith, C. J., Bauer, S. E., Forster, P. M., Fuglestvedt, J. S., et al. (2018). Climate impacts from a removal of anthropogenic aerosol emissions. *Geophysical Research Letters*, 45, 1020–1029. https://doi.org/10.1002/2017GL076079

Schleussner, C.-F., Lissner, T. K., Fischer, E. M., Wohland, J., Perrette, M., Golly, A., et al. (2016). Differential climate impacts for policy-relevant limits to global warming: the case of 1.5 °C and 2 °C. *Earth System Dynamics*, 7, 327–351.

Schleussner, C.-F., Rogelj, J., Schaeffer, M., Lissner, T., Licker, R., Fischer, E. M., et al. (2016). Science and policy characteristics of the Paris Agreement temperature goal. *Nature Climate Change*, 6(9), 827–835. https://doi.org/10.1038/nclimate3096

Seneviratne, S. I., Donat, M. G., Pitman, A. J., Knutti, R., & Wilby, R. L. (2016). Allowable CO2 emissions based on regional and impact-related climate targets. *Nature*, 529(7587), 477–483. https://doi.org/10.1038/nature16542

Sherwood, S. C., & Huber, M. (2010). An adaptability limit to climate change due to heat stress. Proceedings of the National Academy of Sciences of the United States of America, 107(21), 9552–9555. https://doi.org/10.1073/pnas.0913352107

Sillmann, J., Kharin, V. V., Zwiers, F. W., Zhang, X., & Bronaugh, D. (2013). Climate extremes indices in the CMIP5 multimodel ensemble: Part 2. Future climate projections. *Journal of Geophysical Research: Atmospheres, 118,* 2473–2493. https://doi.org/10.1002/jgrd.50188

Stott, P. (2016). How climate change affects extreme weather events. *Science*, 352(6293), 1517–1518. https://doi.org/10.1126/science.aaf7271
Stott, P. A., Christidis, N., Otto, F. E. L., Sun, Y., Vanderlinden, J.-P., van Oldenborgh, G. J., et al. (2016). Attribution of extreme weather and climate-related events. *Wiley Interdisciplinary Reviews: Climate Change*, 7(1), 23–41. https://doi.org/10.1002/wcc.380

Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society, 93(4), 485–498. https://doi.org/10.1175/BAMS-D-11-00094.1

Tobías, A., Armstrong, B., & Gasparrini, A. (2017). Brief report: Investigating uncertainty in the minimum mortality temperature: Methods and application to 52 Spanish cities. *Epidemiology (Cambridge, Mass.)*, 28, 72–76.

- Wang, Z., Lin, L., Zhang, X., Zhang, H., Liu, L., & Xu, Y. (2017). Scenario dependence of future changes in climate extremes under 1.5 °C and 2 °C global warming. *Scientific Reports*, 7, 46432. https://doi.org/10.1038/srep46432
- Warszawski, L., Frieler, K., Huber, V., Piontek, F., Serdeczny, O., & Schewe, J. (2014). The Inter-Sectoral Impact Model Intercomparison Project (ISI-MIP): Project framework. *Proceedings of the National Academy of Sciences of the United States of America*, 111(9), 3228–3232. https://doi. org/10.1073/pnas.1312330110
- Wartenburger, R., Hirschi, M., Donat, M. G., Greve, P., Pitman, A. J., & Seneviratne, S. I. (2017). Changes in regional climate extremes as a function of global mean temperature: An interactive plotting framework. *Geoscientific Model Development*, *10*(9), 3609–3634. https://doi. org/10.5194/gmd-10-3609-2017
- Williams, J. W., Jackson, S. T., & Kutzbach, J. E. (2007). Projected distributions of novel and disappearing climates by 2100 AD. Proceedings of the National Academy of Sciences of the United States of America, 104(14), 5738–5742. https://doi.org/10.1073/pnas.0606292104